DISCUSSION PAPER SERIES

No. 8415

WATER NATIONALIZATION: NETWORK ACCESS, WATER QUALITY, AND HEALTH OUTCOMES

Fernando Borraz, Nicolás González and Marcelo Olarreaga

INTERNATIONAL TRADE AND REGIONAL ECONOMICS



Centre for Economic Policy Research

www.cepr.org

www.cepr.org/pubs/dps/DP8415.asp

Available online at:

WATER NATIONALIZATION: NETWORK ACCESS, WATER QUALITY, AND HEALTH OUTCOMES

Fernando Borraz, Central Bank of Uruguay and FCS-UDELAR Nicolás González, Universidad de Montevideo Marcelo Olarreaga, University of Geneva

Discussion Paper No. 8415 June 2011

Centre for Economic Policy Research 77 Bastwick Street, London EC1V 3PZ, UK Tel: (44 20) 7183 8801, Fax: (44 20) 7183 8820 Email: cepr@cepr.org, Website: www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programme in **INTERNATIONAL TRADE AND REGIONAL ECONOMICS**. Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as an educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Copyright: Fernando Borraz, Nicolás González and Marcelo Olarreaga

CEPR Discussion Paper No. 8415

June 2011

ABSTRACT

Water Nationalization: network access, water quality, and health outcomes*

In the case of natural monopolies there tends to be a trade-off between a higher quality of output provided by private firms, and a better access for poor consumers provided by public firms. This is partly the reflection of differences in objectives by private and public firms. The former tend to be profit-driven, whereas the latter tend to base decisions on political agendas (Chong and Lopez de Silanes, 2005). The objective of this paper is to explore the impact on network access, water quality, and health outcomes of Uruguay's nationalization of water services. An important advantage of focusing on nationalization rather than privatization is that it avoids selection bias due to cherry-picking by firms or governments at the time of privatization. Indeed, nationalization in Uruguay affected all private firms, as water was declared "part of the public domain". Results suggest that the change in ownership led to an increase in the sanitation rate, as well as improvements in water quality. It was also accompanied by a decline in water-related child mortality, although this latter effect tends not to be statistically significant across most specifications.

JEL Classification: D60, F10, H51, I10, L33 and O12 Keywords: access to sanitation, child mortality, nationalization and water quality

Fernando Borraz Banco Central del Uruguay Cerrito Montevideo URUGUAY	Nicolás González Universidad de Montevideo 2440 Dr. Prudencio De Pena Montevideo 11600 URUGUAY
Email: fborraz@bcu.gub.uy	Email: ngonzalez@correo.um.edu.uy
For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=173614	For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=173615

Marcelo Olarreaga Departement d'Economie Politique University of Geneva 40 bld du Pont d'Arve 1211 Geneva 4 SWITZERLAND

Email: marcelo.olarreaga@unige.ch

For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=134114

* We are grateful to María del Carmen Paparamborda and Leticia Rodriguez from the statistics department of the Ministry of Public Health for providing us with access to data on child mortality, as well as Alvaro Capandequy from the energy and water services regulator (URSEA) for providing us with access to data on the quality of water, and the Elections Court secretariat for providing us with data on the results of presidential elections and the water nationalization referendum. We are also grateful to Dale Whittington, Guido Porto, Juan Robalino, Roxana Barrantes and Cristian Rodriguez, as well as seminar participants at a LACEEP workshop and the Geneva Trade and Development Workshop for their very constructive comments and suggestions. The views expressed here are those of the authors and do not reflect necessarily the views of the Banco Central del Uruguay. This work was carried out with the aid of a grant from the Latin American and Caribbean Environmental Economics Program (LACEEP). All errors are our responsibility.

Submitted 12 May 2011

1 Introduction

Back in the 1970s the role of the government as provider of basic services in sectors with a natural monopoly component, such as water supply and sanitation, was hardly questioned. Indeed, it was thought that private firms were likely to abuse their monopoly power in this type of markets, and concentrate their service supply on rich households, leaving poor households without access to a basic service.¹ On the other hand, public companies, because they often tend to be partly driven by a political agenda, would have incentives to ensure access to a maximum of potential voters (at least in democracies).

By the late 1980s the weak economic performance and low productivity of many public companies around the world changed this view (World Bank, 2004). Bad management practices, due to political agendas rather than profit-oriented motives, shed light on large inefficiencies and poor quality of services being provided by these public companies. In the 1990s, privatization of water services was seen by some as a potential solution to the poor performance of publicly owned water monopolies that left more than 1 billion people in developing countries without any access to clean and safe water, and 40 percent of the world's population without access to safe and clean sanitation services (Segerfledt, 2005). UNDP in its Human Development Report of 2006 notes that "not having access to water and sanitation is a polite euphemism for a form of deprivation that threatens life." And Galiani *et al.* (2005) provide empirical evidence that a move towards privatization of water services in Argentina in the 1990s led to a faster decline in child mortality.

France was an early example of a country with privately provided water services. And throughout the 1990s many countries engaged in privatization of water services, starting with England in 1989, followed by Eastern European and Latin American countries. A few Asian and African countries followed them in the mid and late 1990s (Hall *et al.* 2010). Nevertheless, the share of public water companies is still very large.² Moreover, it has

 $^{^{1}}$ In September 2010, the United Nation's Human Rights' Council adopted Resolution 64/292 which declares access to water and sanitation a human right.

²According to Hall *et al.* (2010) water services are owned and run by the public sector in 90% of the largest 400 cities in the world. This figure should be compared to the share of formal employment in public companies across all sectors, which is on average 5 percent according to Kikeri (1999).

increased in the last ten years as backlashes to privatization occurred in many countries.

Uruguay is a recent example of this reversal. An amendment to the Uruguayan constitution was passed in 2004 declaring water as "part of the public domain". The private provision of water was made illegal.³ This led to the nationalization of two water and sewage suppliers in the Department of Maldonado that were privatized in 1993 and 2000. The private water companies that operated in the Department of Canelones since the 1940s also came under the control of the public monopoly.

The apparent reasons for Uruguay's nationalization of water companies were no different from the ones observed in other Latin American countries in the last decade (Bolivia, Argentina, Brazil, etc). Privatization of water services did not deliver on its promises.⁴ Private companies became deeply unpopular due to perceptions, at least, of low or falling quality of water, as well as the high prices private companies charged. A series of highly publicized episodes of low quality water supplied by Uragua and Aguas de la Costa (subsidiaries of the Spanish water companies Aguas de Barcelona and Aquas de Bilbao) led as early as 2003 (in the middle of a financial crisis) to an also well publicized request by the then Minister of Economics and Finance, Alejandro Atchugarry, for Uragua to leave the country.

Whether public or private provision of water leads to better access, quality and health outcomes is an empirical question. The objective of this paper is to explore the impact that the nationalization of water services had on the quality of water (microbiological and inorganic tests) and access to sanitation networks (percentage of households with water-sealed toilets connected to sewer lines) in Uruguay, as well as on an important health outcome: child mortality. The study of Uruguay's nationalization episode is interesting because it avoids the inherently selection bias associated with cherry-picking when looking at privatization. Indeed, nationalization affected all companies as private provision of water was made illegal. Thus, as long as the determinants of the decision to privatize are not serially correlated we would have a reliable identification strategy. Second, household access

³In the Netherlands water privatization was also made illegal.

⁴The reason for the failure of privatization is not necessarily inherent to privatization, but may also be explained by poorly designed contracts (in terms of required investments) or non-adequate regulatory bodies. These are also often associated with problems of corruption (see Chong and Lopez-de-Silanes, 2005).

to Uruguay's piped sewerage network remains particularly low for countries at similar levels of development. The access rate below 50 percent compares badly with other Latin American countries such as Chile, Colombia, Mexico and some comparable Brazilian states in terms of income per capita.

The existing empirical evidence on the impact of privatization on water quality, access and child mortality tends to suggest a positive impact, although there are a few studies suggesting otherwise. Wallsten and Kosec (2005) analyze the effect of water ownership on water quality, measured as the number of violations of the Safe Drinking Water Act (SDWA) in the US between 1997 and 2003. Using panel data at the community level, and controlling for community fixed effects, they find that ownership does not matter in terms of compliance with the SDWA. This may be due to the fact that the study is undertaken in a developed country with high levels of income and therefore a strong demand for high quality water.⁵

In a developing country study, Barrera-Osorio *et al.* (2009) using a difference-in-difference methodology assess the impact of water privatization in Colombia on several outcomes like coverage (percentage of households connected to water service) and water quality (frequency of the service and aspect of water, such as its color). They find that, in urban areas, water access increases and that water quality improves as a result of privatization, while negative effects on access where detected in rural areas. This is consistent with the idea that as water services are privatized the poorest consumers may be the ones left behind.

Regarding the impact on child mortality, Shi (2000) finds a positive effect of private provision of water on child mortality using a cross city model. However, problems of selection bias and endogeneity were difficult to address in a cross section setup. Using a panel data setup, Galiani, Gertler and Schargrodsky (2005) provide convincing evidence that in Argentinean municipalities where water services were privatized, the incidence of child mortality from water-related diseases declined significantly (whereas the incidence of child mortality for other reasons remained stagnant). They therefore provide indirect evidence of improvements in water quality and access. Note, however, that the same critiques towards private water suppliers that were present in Uruguay in the late 1990s and early 2000s were

 $^{{}^{5}}$ In their review of the literature Nauges and D. Whittington (2009) suggest that income elasticities of demand oscillate between 0.1 and 0.4.

present also in the Argentinean press at the time (high prices, water provided by private companies being unfit for human consumption, or the fact that these private companies only honor half of their investment commitments). Galdo and Bertha (2005) also found that water access had positive effects on child mortality in Brazil and Ecuador.

The empirical methodology we followed is similar to the one in Galiani, Gertler, and Schargrodosky (2005). Using panel data around the nationalization episode, we identify differences in sanitation rate, water quality indicators and water-related child mortality between regions that first privatized their water suppliers, and later nationalized them, and those that did not using a difference-in-difference estimator.

Our results suggest that the nationalization of water services had a positive and statistically significant impact on access to the sanitation network, in particular among household in the bottom 25 percent of the income distribution. At the top of the income distribution, only households in Maldonado seemed to experience an increase in access to the sanitation network. But even in Maldonado the impact of nationalization is clearly pro-poor with a larger impact on poor households (coefficients tend to be 2 times larger and statistically different from the coefficient in the rich household sub-sample).

Nationalization seemed to have led to higher quality of water as well. Indeed, the impact of nationalization on the detection of abnormal levels in microbiological and inorganic water tests is always negative and has a relatively large coefficient, suggesting that nationalization led to an improvement of water quality.

Finally, our different specifications tend to suggest that water nationalization had little impact on child mortality, even when we disentangle water and non-water related diseases. Indeed, in most cases the coefficients on nationalization of water services is not statistically different from zero, even though in most cases it has a negative sign, suggesting that child mortality declined after nationalization.

To summarize, the nationalization of water services in Uruguay suggests that the public sector can do it as well or better than the private sector. This goes against most of the existing evidence for developing countries that generally shows that privatization leads to higher levels of sewerage access and water quality, as well as lower levels of child mortality. This could be partly explained by a badly implemented privatization in Uruguay that was not accompanied by adequate regulation and institutions. Also, it is possible that Uruguay's public company is more efficient in terms of access, water quality and public health objectives than public water companies in other countries. Overall, the empirical results in this paper suggest that privatization is not the silver bullet that could lead the way towards better access to clean and safe water.⁶

The reminder of the paper is organized as follows. Section 2 describes the functioning of the water system in Uruguay. Section 3 discusses the empirical methodology, and Section 4 presents the data and some descriptive statistics. Section 5 presents the results, and section 6 concludes.

2 The Water System in Uruguay

At the turn of this century water and sewage service in Uruguay were provided by fifteen private companies and a publicly-owned company called OSE (Obras Sanitarias del Estado). OSE had an exclusive monopoly in water provision in Montevideo (Uruguay's capital city) and almost all of the national territory accounting for more than 90% of the connections. The private sector was basically composed of three big companies (Uragua, Aguas Corrientes El Pinar, and Aguas de la Costa) and approximately twelve water providers with less than 500 connections each. These small providers were cooperatives founded by residents to provide water services in areas that OSE was not reaching or was reaching with poor quality water.

Uragua started to provide water and sewage service in urban and sub-urban areas of Maldonado in 2000. Maldonado has 150,000 inhabitants of a total of 3,300,000 in Uruguay. Uragua served the west of the Maldonado stream with the exception of the city of Aiguá. More specifically, Uragua served the capital of Maldonado (50,417 inhabitants), San Carlos (23,878), Pan de Azúcar (6,969), Piriápolis (7,579), Cerros Azules, Nueva Carrara, Pueblo Gerona, West of River Solís, Silver River, Highway 9 North, and the international well-known

⁶Note that our result could also be explained by an initial euphoria and push to better management immediately after the nationalization that could eventually vanish as time goes by.

resort of Punta del Este $(7,298 \text{ inhabitants})^7$.

Uragua was owned by Aguas de Bilbao, a Spanish water provider. In 2004 Uragua entered in litigation with OSE because of a breach of contract, following some well publicized episodes of colored water in Maldonado. Following the 2004 amendment to the Uruguayan constitution,⁸ Urugua reached a deal with the Uruguayan government and all Uragua assets were transferred to OSE by the end of 2005.

Aguas Corrientes El Pinar was created in 1946, six years before the creation of OSE, with the objectives of fostering land sales in El Pinar area which legally required connections to water and sewage services. It served less than 1,000 clients until the 1990s, but by the time of the nationalization it had reached around 5,000 clients due to the rapid development of the area over the last decade. Interestingly, only 12 persons worked full time in Aguas Corrientes El Pinar. The connection fee was around USD 200 and the average consumption was USD 10.

Aguas Corrientes el Pinar provided services in El Pinar, but also the adjacent areas of San Cristóbal, El Palmar, Médanos de Solymar, Lomas de Solymar, Autodromo, and Colinas de Solymar in the department of Canelones. Canelones is the second Department in Uruguay in terms of population size with 500,000 inhabitants. The population in El Pinar and Lomás de Solymar is around 33,000. Aguas Corrientes el Pinar was nationalized in December 2006 and its assets were also transferred to OSE.

The third biggest private company in Uruguay was Aguas de la Costa. It was a joint-venture between a local company, S.T.A Ingenerios and Benecio S.A., which had around 10% ownership, and the Spanish company Aguas de Barcelona, a subsidiary of Suez Lyonaisse des Eaux, which owned the rest of the company. Aguas de la Costa signed the concession in 1993 for twenty five years. The company supplied water and sewage in the wealthier areas of La Barra, Manantiales and José Ignacio, which is next to Punta del Este. This area is to the east of the Maldonado stream. The company had around 3,100 customers and the connection fee for water and sewage was USD 2,000 and USD 1,000 for water only. A special rate for

 $^{^{7}}$ We take all year around population and not the tourist population, which can reach hundreds of thousands during the summer.

 $^{^{8}}$ The amendment that went through a referendum declared water as being part of the public domain.

low income households was available. The average water consumption ranged between USD 40 and USD 100. After an agreement with the government, the company left the country in 2005. Aguas de la Costa assets were transferred to OSE in 2005.

Figure 1 provides a map of Uruguay and Figure 2 provides maps of Canelones and Maldonado to better understand the location of the privately-owned water companies. The left panel of Figure 2 shows the department of Canelones by judicial section. The judicial section 19 in Canelones was served principally by Aguas Corrientes el Pinar. The rest of the judicial sections of Canelones were operated by the publicly-owned OSE. The right panel of Figure 2 shows the map of the Department of Maldonado by judicial section. The judicial sections 1,2, 3 and 5 were served by Uragua. The judicial section 6 was operated by Aguas de la Costa. Finally, judicial section 4, which contains the city of Aigua, was not privatized and was controlled by the public company OSE.

It is clear that Maldonado was particularly hit by the nationalization reversal. At the turn of the century, the only city in Maldonado served by the publicly-owned OSE was Aigua with 2,676 inhabitants. By 2006, OSE was the only provider of water services in Maldonado.

Table 1 shows the evolution of some indicators of the public company OSE between 2002 and 2006. For instance, the number of employees did not suffer a substantial modification after nationalization. As it is expected, the volume of water produced grew after nationalization since the water service coverage increased. Moreover, the population served with sewerage rose from 531,300 in 2004 to 729,100 in 2006. Total gross fixed assets (including work in progress) increased sharply (around 38%) between 2004 and 2006. This partly includes acquisitions of the private companies, but also new infrastructure investments through a sanitation project supported by the World Bank after nationalization. Finally, contrary to what could have been expected, average prices charged by OSE increased after nationalization. The fixed charge per month for water and wastewater services for residential customers increased by approximately USD 1 after nationalization. Connection fees also increased.

3 Methodology

The ideal methodology to assess the impact of privatization (nationalization) on different outcomes would be an experiment involving a random assignments to treated and non-treated individuals. That is, we would have cities which are randomly chosen to be privatized, usually called the "treatment group" and other which are not (the "control group"). In our case privatization was not carried out at random and therefore we need to find out an alternative methodology which mimics as best as possible a natural experiment. We followed the existing literature and adopt a quasi-experimental difference-in-difference methodology.⁹

The identification strategy we followed is similar to the one in Galiani *et al.* (2005) in where they search for systematic differences in changes in child mortality rates between regions that have privatized and those that have not changed the ownership structure of water companies using a difference-in-difference approach.¹⁰

Selection bias due to cherry-picking at the moment of privatization is an issue in this setup, that they address through different methods. In our case, given that the nationalization was imposed on all private firms allows us to circumvent this problem, as long as the determinants of privatization are not serially correlated. In the case of Uruguay's nationalization of water companies in 2005-2006, the condition seems to be satisfied as some of the private companies in Canelones were funded as early as 1946, and it is therefore unlikely that the determinants of "privatization" at that time are correlated with the decision to nationalize all water companies in 2005-2006.

On the other hand, in Maldonado the privatization episode was more recent. Aguas de la Costa and Uragua were originally privatized in 1993 and 2000 respectively. In that sense, one can argue that there was selection in the privatization decision at the time. If this latter argument applies, these endogenous determinants of privatization may also be biasing the estimates of the impact of nationalization on the outcome variable.

⁹For a randomized experiment of spring water cleaning in rural Kenya, see Kremer, Leino, Miguel and Zane, (2009).

¹⁰Lee *et al.* (1997) show that this reduced form approach may downward bias the effects of treatment as households adjust their behavior to the new environment, and provide an alternative structural approach to estimate the impact of better quality on health outcomes.

For example, suppose that the government decide to privatize companies in judicial districts that were more profitable and had better prospects in order to maximize the short-run financial benefits from the privatization. If profitability positively affects performance over time (i.e., performance is serially correlated) we will observe that nationalization had led to a better performing water company, but this was just due to the trend in the performance of water companies in that region. In this context, even if nationalization adversely affected performance, estimation might not identify this effect because the treatment group includes a disproportionate number of utilities that perform well.

There are two answers to this. First, as observed in the Data section, privatized and public water companies in Maldonado had similar trends in performance before the nationalization of water companies. And in some of the difference-in-difference specification we control for general trends using year fixed effects, as well as jurisdiction-specific trends. Second, the decision to nationalize water companies was mainly politically as illustrated in Table 2 which shows data from the Electoral Court regarding the water referendum and the national election of 2004, as well as the department elections in 2005. There is a strong correlation between support for the water referendum and support for the left wing party. For example, if we considered the Department of Maldonado, which had two private water providers, we observe that the referendum was defeated only in cities where the national or department elections went for right wing parties (Pan de Azúcar, Punta del Este and Aigua). One therefore suspects that the decision to privatize in the 1990s was also politically and not performance driven.

Our general econometric model is given by:

(1)
$$y_{it} = \phi D_{it} + x'_{it}\beta + \lambda_t + \alpha_i + u_{it}$$

where y_{it} is the outcome variable in city (or judicial section) *i* in year *t*. We consider three different outcome variables: sanitation rates, water quality, and child mortality. The unit of observation *i* are cities in the case of sanitation and water quality analysis, and judicial

sections in the case of child mortality. D_{it} is a dummy variable indicating that the water company in city (or judicial section) *i* in period *t* is publicly-owned, α_i is a city (or judicial section) specific fixed effect, λ_t is a year effect, x_{it} is a vector of control variables and u_{it} is a city (or judicial section) time-varying error (distributed independently across department and time). The regression is structured to capture the effects of aggregate factors and department specific responses to aggregate factors.

The parameter of interest is ϕ which measures the casual impact of the type of ownership (public vs. private) of the water company on water quality. When sanitation rate is the outcome variable, a positive coefficient will indicate that nationalization of private companies led to larger sanitation rates conditional on year and city fixed effects, as well as control variables at the city level. On the other hand, when considering water quality tests (abnormal levels of microbiological and organoleptic elements) or child mortality rates as the dependent variable, a positive coefficient implies that nationalization led to a higher number of tests with abnormal results and more child mortality, again conditional on year and city (or judicial section) fixed effects, and other control variables at the city (or judicial section) level.

In order to address the problem that different departments may have been following different trends before nationalization, we include a city (or judicial section) specific time trend. This will correct for the fact that nationalization may be correlated across time with other determinants of our outcome variables that are unrelated to the change in ownership of water companies. If we do not control for this city (or judicial section) specific time trends, our estimates may be biased due to a spurious correlation between nationalization of water companies and the three different outcome variables. Formally,

(2)
$$y_{it} = \phi D_{it} + x'_{it}\beta + \lambda_t + \gamma_i trend + \alpha_i + u_{it}$$

where γ_i trend is a city (judicial section) specific time trend.

Another important issue is that in panel data models observations tend to be correlated across time within individual city or judicial sections. One possible solution to address the serial correlation problem is to use robust standard errors clustered at department-section judicial level. In this context, asymptotic statistical inference depends on the number of cluster and time periods. A small number of clusters could result in biased (clustered) standard errors, tending to underestimate inference precision.

Bertrand, Duflo and Mullainthan (2004) analyzed the performance of different alternative solutions to the serial correlation problem: 1) parametric methods, that is, specifying an autocorrelation structure; 2) block bootstrap; 3) "ignoring time series information", that is, averaging the data before and after treatment; and 4) using an "empirical variance-covariance matrix". Bertrand, *et al.* find that the latter perform better than the others, but it does not work properly with small sample size. Moreover, Cameron, Gelbach and Miller (2008) states that block bootstrap work properly with small number of groups.¹¹

In our case, bootstrap clustered standard errors are similar to clustered standard errors and in most cases the latter method report greater standard errors and in some cases conventional standard error are larger than the latter. As a conservative criterion, we decided to report the method with the largest standard errors in each case. Depending on the nature of the outcome variable: continuous, count or fractional, we will use different estimators that will be discussed in the results section.

4 Data sources and variable construction

We start describing data sources and variable construction for the analysis of the impact of nationalization on access to water sanitation networks. We then turn to data sources and variable construction for the study of the impact of nationalization on water quality, and we finally discuss data sources when we use child mortality as the outcome variable.

4.1 Access to sanitation networks

The data regarding the percentage of households with water-sealed toilets connected to sewer lines are obtained from the annual Uruguayan national household survey, *Encuesta*

¹¹An alternative solution is to use the method proposed by Bell and McCaffrey (2002) "bias correction of clustered standard errors", but unfortunately this approach cannot be applied in a difference-in-difference set up.

Continua de Hogares (ECH) conducted by the National Statistical Office of Uruguay, *Instituto Nacional de Estadística* (INE). The ECH is the main source of socio-economic information about Uruguayan households and their members at the national level. The surveys were carried out throughout the year with the objective of generating a proper description of the socio-economic situation of the entire population.

The ECH also includes questions about household living conditions. In particular, the survey asks whether water-sealed toilets are connected to sewer lines. Hence, we generate a dummy variable that takes the value 1 if the household is connected to sewer lines and 0 otherwise. Then, we aggregate the data by city in order to obtain the percentage of households with sanitation access in each city. We therefore work with panel data by city from 1986 to 2009. The time-span includes the pre-privatization period in the case of Maldonado, as the two privatizations in Maldonado occured in 1993 and 2000 as discussed earlier.

The ECH survey is only representative at the Department level or at the city level for largest cities in terms of population. Therefore, and to have a representative sample, we keep only capital cities of the different Departments and other big cities in our sample. We have a total of 35 cities, three in the treatment group (Maldonado, Pan de Azúcar and San Carlos) and 32 in the control group: Artigas, Bella Unión, Canelones, Carmelo, Colonia, Dolores, Durazno, Florida, Fray Bentos, Lascano, Libertad, Melo, Mercedes, Minas, Montevideo, Paso de los Toros, Paysandú, Periferia Canelones, Rivera, Rocha, Rosario, Río Branco, Salto, San José de Mayo, San Ramón, Santa Lucía, Sarandí del Yí, Sarandí Grande, Tacuarembó, Tranqueras, Treinta y Tres, Trinidad and Young.

Due to the fact that there are some selected cities which were not surveyed in some years (mainly in the ECH older edition), we have an unbalanced panel which could possibly imply panel attrition bias. For instance, in the treatment group, of the total 24 time periods, Pan de Azúcar appears 15 times and San Carlos 18 times. In the control group, we have observations for Lascano in 10 of the 24 time periods, and for Bella Unión, Libertad and Rosario in 12 years, Santa Lucía in 15 years, Carmelo, Dolores, Paso de los Toros, Río Branco, San Ramón, Sarandí del Yí and Sarandí Grande in 17 years, and Young in 18 years. So, we have a total of 735 observations¹². Some cities started to appear in the ECH because of their rapid growth in terms of population and therefore we checked the robustness of results to a smaller sub-sample in terms of time-span: 1993-2009.

The are two observations that need to be made. First, control cities largely exceed treatment cities and hence we will also estimate our model reducing the number of controls to only capital cities. The sanitation data of each capital city are available in every year of the whole period, so in this case panel attrition problems are potentially solved. Second, since we lose observation as we drop cities, small sample bias could arise. Thus, this could be seen as a trade-off between small sample bias and possible panel attrition bias, which provides some robustness checks to our results.

In Figure 3, we present the evolution of the average sanitation rate of the 35 cities. Overall, we can see that it has an upward trend and in recent years it grows at a faster pace which leaves sanitation above its trend level. In the econometric analysis we will include general trends in network evacuation rates to control for this. Note that the water utilities of Maldonado, Pan de Azúcar and San Carlos belonged to the private company Uragua between 2000 and 2005. Thus, we have for those cities a pre-privatized period 1986-1999, in where water utilities were serviced by OSE the public company, and the period after nationalization, 2006-2009, when the water utilities returned to OSE.

Figure 4 illustrates the evolution of the average sanitation rate for treated (solid line) and non-treated cities (dash line). There seems to be a higher variance for the treatment group, but before privatization the trend in network evacuation rates across the treatment group and the control group were similar, especially between 1997 and 2000, the period immediately before privatization. During the privatization period between 2000 and 2006, the rate of growth in sanitation access in treated cities seemed to accelerate relative to non-treated cities. And this trend accelerates again after nationalization in 2006.

Figure 5 plots the evolution of sanitation rates for each city belonging to the treatment group (Maldonado, Pan de Azúcar and San Carlos), as well as Ciudad de la Costa, which used to be provided by the private-owned company Aguas Corrientes El Pinar. We observe

¹²There are 35 cities. We have 21 cities times 24 (1986-2009 period) which represent 504 observations. In addition, we have $1 \ge 10 + 3 \ge 12 + 2 \ge 15 + 7 \ge 17 + 2 \ge 18 = 231$ observations.

some interesting variability in this indicator of performance. Access to the sanitation network in Ciudad de la Costa was nearly zero until 2005, when the nationalization process begun and after that, we observe an upward trend. Ciudad de la Costa was not included in the sample because, as we can see in Figure 5, its sanitation rate is far from the average of the rest of the cities of the treatment group (and from the overall average) and thus it could interfere in the estimation of a reliable counterfactual.

Table 3 provides some descriptive statistics by treatment and control groups. Overall, the treatment and control group present similar characteristics on average. Nevertheless, the treatment group has a smaller proportion of poor and extremely poor inhabitants, which should be captured by our city specific effects in the econometric analysis. Indeed, using a difference-in-difference method, we will try to answer whether the higher level of access to the sanitation network observed in the treatment group after nationalization can be attributable to a causal effect. This evolution could be following a trend and thus in some specifications we include time trend by city as a way to check if our results are robust.

4.2 Water quality tests

Data on water quality tests comes from the energy and water services regulator (URSEA). In 2004, URSEA jointly with the chemistry department of University of La República created a water quality unit (UAA) which is in charge of monitoring the water supply system in the whole country, according to the guidelines of the World Health Organization. This unit carries out several water tests (in the distribution network) to measure the water quality provided to consumers.

Our units of observation are cities from Uruguay's 19 Departments. We have data for the period 2004-2009 but some observations are missing for some of the cities. As treatment cities we have Maldonado, Pan de Azúcar, Piriápolis, Punta Ballena, Punta del Este and San Carlos. And as control cities: Artigas, Canelones, Colonia, Dolores, Durazno, Florida, Fray Bentos, La Paloma, La Paz, Las Piedras, Melo, Mercedes, Minas, Montevideo, Pando, Paysandú, Progreso, Rivera, Rocha, Salto, San José, Atlantida, Tacuarembó, Toledo, Treinta y Tres and Trinidad. We have data for two different microbiological tests and two organoleptic tests, which we will use as outcome variables because of their importance in terms of direct and indirect negative effects on health. The two microbiological tests are for fecal coliforms, and pseudomonas aeruginosa. The two organoleptic tests are ph and cloudiness tests.

The first test indicates whether the fecal coliforms exceed the accepted limit value. The second test indicates whether pseudomononas aeruginosa is present. In the case of organolpetic tests, the UAA reports the observed value. In the case of ph tests, the upper limit is 8 mg/Land, and higher levels are considered abnormal. The upper limit in the cloudiness test is 5, so a result greater or equal to this value represents a high level of cloudiness in the water, which could indirectly affect health via a higher likelihood of bacteria formation and a reduction in the quantity consumed of water.

The outcome variable that captures abnormal levels of microbiological or organoleptic substances is constructed as follows. For each test we generate a dummy variable that takes the value 1 if the test is above the accepted limit and zero otherwise. We then sum these four binary variables to create a count variable that measures the number of tests that showed abnormal levels in each city.

One possible drawback of this data is that tests have become better and more precise over time, allowing to detect more often abnormal levels of substances. This could bias our estimates, as we would observe a deterioration of the water quality throughout the period due to better testing techniques, rather than water quality. This is addressed in our econometric framework by the use of year fixed effects as control variables.

Figure 6 plots the evolution of the average of this count variable for treatment and control groups. We actually observe initially a downward trend in both treated and control groups until nationalization. And the average of the treated group declines faster. After nationalization, the treatment group experience almost no detection in abnormal levels until 2008, whereas the control group had a slight increase in the detection of abnormal levels. From 2008 there is an increase in the detection in both the treatment and the control group. This could be explained by the increase in the detection capacity as mentioned above.

In Table 4, we provide some descriptive statistics by treatment and control group.

Overall, and as we previously observed in the figures, the treatment group presents a better level of quality if we consider the entire period.

4.3 Child mortality

The data on child mortality by judicial section and cause of death come from the Ministry of Public Health (Dirección General de la Salud-Departamento de Información Poblacional-Estadísticas Vitales). An important characteristic of Uruguay's health infrastructure is that it tends to be concentrated in its capital: Montevideo. Children's health is no exception, and when children from all over the country need serious care, they are sent to the Pereira Rossell in Montevideo, which is the largest children hospital in Uruguay. If the child eventually dies in the Pereira Rossell, public mortality statistics will report this death as occurring in Montevideo. In order to avoid this bias we asked the Ministry of Public Health to provide us with information regarding the mother's residence by judicial section, rather than the location of the child's death.

Another interesting aspect of the dataset is that it contains detailed information on the causes of death. Galiani *et al.* (2005) lump together all death caused by infectious and parasite diseases and perinatal death as being water-related. Our dataset includes information taken from the death certificate signed by the Judicial Officer, which identifies the cause of death in great detail. This allows us to more precisely identify whether the type of infectious disease and exclude child deaths which were caused by infectious diseases that are not water-related.¹³ This could be done with relatively more precision in the case of non-perinatal deaths, i.e., those that occur between 5 weeks and 1 year old. In the case of perinatal deaths, which occur within the first 4 weeks it is often difficult to identify the exact cause of death.

The evolution of total, perinatal, and non-perinatal child mortality in Uruguay around

¹³We include the following causes of death as being water-related: myotonic disorders, bacterial sepsis of newborn, unspecified, necrotizing enterocolitis fetus and necrotizing enterocolitis of the new birth septicemia, unspecified, meningitis, unspecified encephalitis, myelitis and encephalomyelitis, unspecified, bacterial meningitis, unspecified, congenital viral hepatitis, sepsis of new birth by other staphylococcus, and other unspecified, septicemia due to candida.

the nationalization period is shown in Figure 7.¹⁴ Child mortality experienced a downward trend between 2003 and 2006, with an important increase in 2007, followed by a decline in 2008 and 2009, which leaves child mortality below its trend level. It is interesting to note that during all this period the average GDP growth was 6.3% and ranged from 2.2% in 2003 to 11.8% in 2004.

Figure 8 shows the evolution of child mortality rates by water-related diseases for treated and control groups separating the regions affected by the privatization in Canelones and Maldonado of Lavalleja and Rocha (and we also include the part of Canelones which always belonged to the public company). In both cases, we observe a downward trend through most of the period. But there are some interesting differences before and after nationalization: 1) before nationalization water-related child mortality has a similar trend across the treatment and control Departments, but the treatment group tends to have a higher level of water-related child mortality; 2) after nationalization water-related child mortality declines faster in the treatment group at least until 2009, and the level of child mortality in the treatment group is generally below the level of water-related child mortality in the control group. This fact could be attributable to a causal effect.

In the econometric analysis we work with a panel data at the judicial section level. Our control group are the neighboring judicial districts in Canelones and Maldonado. More precisely, the judicial sections with private water companies located in Maldonado are judicial sections 1 to 3, 5 and 6. We merge judicial sections 3 and 5 because of representativeness of judicial section 3 in terms of population. In Canelones judicial section 19 is subject to treatment. As control groups we will use neighboring judicial sections that did not privatize: judicial section 4 in Maldonado, and judicial sections 1 to 7, 10, 11, 13, 14, 16, 17 and 18 in Canelones. We also use as control judicial section 1 in Maldonado's neighboring Department of Lavalleja and judicial sections 1, 3, 4 and 5 in Maldonado's neighboring Department of Rocha. In total, we have 25 judicial sections in our analysis from 2003 to 2009. Therefore, the total number of observation is 175. In Table 5, we present descriptive statistics.

¹⁴Figures for Uruguay are approximated by figures for the departments of Canelones, Lavalleja, Maldonado and Rocha, which represent our choice of treatment and control groups.

5 Results

We start discussing the results of the estimation of equations (1) and (2) when access to sanitation rates is the outcome variable. We then turn to the estimates obtained when water quality is the outcome variable, and we conclude this section with the results obtained when child mortality is used as the outcome variable.

5.1 Access to sanitation networks

We estimate equations (1) and (2) for different subsamples, with and without control variables. Because the left-hand-side variable is a fractional variable (percent of households with water-sealed toilets connected to sewer lines) we used a Papke and Wooldridge (2006) estimator. Control variables include average completed years of education of the household head and average real per capita household income at the city level, as well as accumulated precipitations at the department level. We expect the three control variables to be positively correlated with network evacuation rates.

Table 6 presents the results. In almost all subsamples, there is a positive and statistically significant impact of public ownership on sanitation rates. A positive effect means that the presence of a public provider (i.e., nationalization) led to a higher access to sanitation networks. The coefficient which captures the causal impact is on average 0.08, which means that nationalization led to an 8% increase in access to sanitation networks.

In order to check whether the increase in access to sanitation networks occurred where we expect it, i.e., among poor households, we aggregate the data at the city level using only the lower 25th household income percentile in one case and the higher 25th in the other. Table 7 and Table 8 provide estimates for these two sub-samples. As expected, in the first case we found a positive, large and statistically significant impact of private ownership on sanitation rate for all sub-samples. The point estimate is larger than the one reported in Table 6. Access to sanitation networks for those at the bottom of the income distribution increased by 15% on average as a result of nationalization. In the case of rich households, we found a positive and statistically significant effects but only for some specific sub-samples. Moreover, the

coefficient is always statistically smaller than the coefficient for poor households. In sum, nationalization has a positive impact on access to sanitation networks, and this effect is stronger for poor households.¹⁵

Because serial correlation may be an issue in this difference-in-difference setup, we parametrically model the serial correlation of the error term as a first autoregressive process AR(1). We use different estimation methods which imply different assumptions. Results are reported in Table 9. In the first column, we report pool feasible generalized least squares estimates, allowing the error term to be correlated over cities, i.e., we allow for error correlation across panels or usually called "spatial correlation". In the second column, error correlation across individuals is assumed to be identically and independently distributed. In the third column, we use the within estimator to estimate fixed effects, and in the last column we provide estimates using Driscoll and Kraay method to obtain Newey West type standard errors, which allow error autocorrelation of the general form. We use the sub-sample, which only includes capital cities apart from Montevideo. Again, in all cases there is a positive impact of nationalization on access to network sanitation.

This increase in sanitation access is consistent with the fact that a USD 50 million loan was obtained from the World Bank for an OSE project on sanitation and residual treatment after nationalization of the private companies.¹⁶ In addition, OSE has published on its website that after nationalization work related to sanitation improvements, which has been stopped since 2002, restarted again. Sanitation projects in Ciudad de la Costa, Punta del Este and Maldonado, where private companies were located became priority in OSE's agenda.

 $^{^{15}}$ We also tried these regression for households above and below the poverty line. The results, which are available upon request, are similar.

¹⁶The World Bank has financially supported Uruguay in the development of the water and sanitation services with 3 investment loans in different time periods: 1) Water Supply Rehabilitation project (1988-1999) - USD 22.3 million; 2) OSE Modernization and Systems Rehabilitation project, APL-1 (2000-2007) – USD 27 million; 3) OSE Modernization and Systems Rehabilitation project, APL-2 (ongoing since 2007) – USD 50 million. Additionally, there were other loans which focus on technical support: Public Services Modernization Technical Assistance project (2001-2008) – USD 6 million. These objective of these loans were to help Uruguay carry out investments in the water sector infrastructure, improving efficiency and coverage of the water supply and sanitation service.

5.2 Water quality tests

The outcome variable is the count of tests that reported an abnormal level of microbiological or organolpectic substances. This type of right-hand-side variables requires an appropriate estimator. We will use a Poisson and a Negative Binomial estimator to take into account over-dispersion in water tests (a non-constant ratio of variance over conditional mean). Because there is a large number of zeros in the data (see Table 4) we also perform a Vuong test which indicated that the zero-inflated Poisson was the appropriate model.

We used as control variables accumulated precipitations, minimum temperature and average temperature at the department level. We expect precipitation to be positively correlated with the number of abnormal quality tests, since a high level of precipitations is likely to negatively affect the functioning of the water network, making water tests more prone to detecting higher levels of undesired substances. Low temperature could increase the likelihood of failure of the network distribution system and a high average temperature could contribute to the reproduction of bacterium like coliforms, so we expect a negative coefficient on the former and a positive coefficient on the latter.

Table 10 reports the estimates with and without control variables. The first two columns provide the zero-inflated Poisson estimates, and the last two columns the zero-inflated Negative Binomial estimates. Control variables have the expected signs across both specifications, but none of the variables is statistically significant. The public provision of water services is always negatively associated with abnormal levels of undesired substances in water quality tests, and the impact is statistically significant at the 10 percent level. It is also very large, with a reduction of 0.7 tests per city showing abnormal levels after nationalization. Thus, results suggests that water quality improved with nationalization.

5.3 Child mortality

Because child mortality by water-related diseases is a count variable, we address the count nature of the data using either a Poisson, or a Negative Binomial estimator to control for over-dispersion in child mortality (a non-constant ratio of variance over conditional mean), when measurement error is likely to be different in judicial districts with different levels of child mortality. There is also a large number of judicial districts that report no child mortality throughout the period and therefore we used a Vuong test to see whether a Poisson or a zero-inflated Poisson model is the appropriate estimator. The results of the Vuong tests suggest that the standard Poisson model is the adequate estimator in spite of the large presence of zeroes. We also provide estimates using child mortality rates as the dependent variable (instead of the count in child mortality). Because child mortality rates (child mortality over the number of births) are a fractional variable we used Papke and Wooldrige (1996) correction.¹⁷

We used as control variables average completed years of education, average household income per capita and the unemployment rate at the judicial section level, as well as annual accumulated precipitations and average minimum temperatures at the department level. The first three variables are expected to be correlated with child mortality. Higher income and education will be negatively correlated with child mortality, as they will help households avoid diseases that may lead to the death of a child. On the other hand, higher levels of unemployment (conditional on income and education) may tend to positively affect child mortality through the additional stress and tension it imposes on households. A high level of accumulated precipitation is associated with floods which affect children's health, and therefore is expected to be positively correlated with child mortality. The variable average minimum temperature is expected to be negatively correlated with child mortality since higher minimum temperatures could be linked with a less risky environment in terms of illnesses.

The results of different panel estimation with and without control variables are provided in Table 11. The first six columns present results using Canelones and Maldonado as the treatment group, and the last six columns results are for Maldonado only. In each case, results are first given for the Poisson estimator, then the negative Binomial estimator and finally for the Papke and Wooldridge (1996) fractional response estimator.

The estimated coefficients capturing the impact of having a public water provider on child mortality are generally negative, but they are statistically significant only for the Papke and

¹⁷Note that in the Poisson and Negative Binomial estimation we also control for the total number of births in each judicial district, and impose a coefficient equal to 1.

Wooldridge estimates, and only when we do not use control variables. There are no major differences between the estimates obtained when using only Maldonado as the treatment group, and Maldonado and Canelones judicial districts as the treatment group.

In sum, the nationalization of water services does not seem to have had an important impact on water-related child mortality. But, if anything, nationalization has led to a reduction, not an increase, in child mortality, as could have been expected from earlier papers. Note that the lack of statistical significance of some of the results on water-related child mortality could be partly explained by the low numbers of water-related child deaths in Uruguay as reported in Table 5.

6 Conclusions

The question of market versus government failures in the provision of water services is complex, and unlikely to be answered without empirical evidence. In this paper we examine the impact of nationalization of water services on access to network and quality of water, as well as a health outcome: child mortality. Contrary to most of the existing literature we identify the impact not through the privatization of public firms, but rather through the nationalization of private firms. This is an interesting advantage of our case study because nationalization suffers less than privatization from selection bias, i.e., the fact that governments and private firms select the public companies that should be privatized. In the case of the nationalization of water services in Uruguay, all existing private companies were nationalized, which help us identify the impact of the reform on the quality of the service provided without having to worry about firms or governments choosing the best (or the worst) firms for privatization.

Several outcomes variables were used: access to sewage network, water quality and water-related child mortality. Using difference-in-difference estimators, we found that Uruguay's nationalization of water services in 2006 led to an improvement in access to the sewage network, as well as an improvement in water quality. We also found evidence of a decline in water-related child mortality, but this latter effect tends not to be statistically significant.

These results contrast with existing evidence for privatization of water services in other Latin American countries, which tend to find that privatization led to a decline in child mortality, and an increase in water access and quality. Future research should try to disentangle the determinants of these two different outcomes to help understand why privatization did not have the expected impact in Uruguay.

Potential explanations for these differences may lay on the type of regulations at the time of privatization or nationalization (investment requested, universal services), the functioning of regulatory bodies, or badly designed contracts and bidding processes, etc. A detailed examination of these differences can help understand what works and what does not work in terms of water privatization. Also differences in the functioning of public companies (external funding, composition of the board of directors, etc) could help explain why in some countries public providers of water services seem to do better than in other countries.

Finally, the results in this paper suggest that the focus on private versus public ownership of natural monopolies such as water providers may be misleading. The institutional environment under which the natural monopoly operates may be much more important. In Uruguay, at least, the public operator provided a service of equal if not better quality than the one previously provided by private firms.

References

Barrera-Osorio, F., Olivera, M. and Ospino, C. (2009). "Does Society Win or Lose as a Result of Privatization? The Case of Water Sector Privatization in Colombia" Economica 76(304), 649-674.

Cebu Team. (1991): "Underlying and Proximate Determinants of Child Health: The Cebu Longitudinal Health and Nutrition Survey", American Journal of Epidemiology 33(2).

Chong, A. and F. Lopez-de-Silanes (2005). "Privatization in Latin America: Myths and realities", Standford University Press.

Fauconnier, I. (1999). "The Privatization of Residential Water Supply and Sanitation Services: Social Equity Issues in the California and International Contexts", Berkeley Planning Journal 13, 37-73.

Foster, V. (1999): "Literature Review for Regional Studies Project on Privatization and Infraestructure Services of the Urban Poor", mimeo, World Bank.

Galdo, G. and Bertha, B. (2005). "Evaluating the Impact on Child Mortality of a Water Supply and Sewerage Expansion in Quito: Is Water Enough?" OVE Working Papers 0105, Inter-American Development Bank, Office of Evaluation and Oversight (OVE).

Galiani, S., Gertler, P. and Schargrodsky, E. (2005) "Water for Life: The Impact of the Privatization of Water Services on Child Mortality" Journal of Political Economy, University of Chicago Press 113(1), 83-120.

Galiani, S., Gertler, P., Schargrodsky, E. and Sturzenegger, F. (2005): "The Benefits and Costs of Privatization in Argentina: A Microeconomic Analysis", in Chong, A. and F. Lopez-de-Silanes (eds.), Privatization in Latin America: Myths and Realities, Standford University Press.

Hall, D., Lobina, E. and Corral, V. (2010). "Replacing failed private water contracts", PSIRU (Public Services International Research Unit), University of Greenwich, London. Heckman, J. H. Ichimura, J. Smith and P. Todd (1998a): "Characterizing Selection Bias using Experimental Data", Econometrica 66(5).

Heckman, J., R. LaLonde and J. Smith (2000): "The Economics and Econometrics of Active Labor Market Programs", in Ashenfelter, O. and D. Card (eds.), The Handbook of Labor Economics, Volume 3A, North-Holland.

Kikeri, S. (1999). "Privatization and labor: what happens to workers when governments divest?" World Bank, World Bank Technical Paper No. 396.

Kremer, M., J. Leino, E. Miguel, and A. Zwane (2009). Spring Cleaning: Rural Water Impacts, Valuation and Property Rights Institutions. NBER working paper #15280.

Lee, L., M. R. Rosenzweig, and M. M. Pitt (1997). "The Effects of Improved Nutrition, Sanitation, and Water Quality on Child Health in High-Mortality Populations", Journal of Econometrics 77.

Merrick, T. W. (1985). "The Effect of Piped Water on Early Childhood Mortality in Urban country-regionplaceBrazil, 1970 to 1976", Demography 22(1).

C. Nauges and D. Whittington (2009). "Estimation of water demand in developing countries: an overview", World Bank Research Observer 25(2), 263-294.

Noll R., M. Shirley and S. Cowan (2000). "Reforming Urban Waster Systems in Developing Countries" in A. Krueger (ed.) Economic Policy Reform: The Second Stage, University of Chicago Press, Chicago.

Papke, L. and J. Wooldridge (1996). "Econometric Methods for Fractional Response Variables with an Application to 401(K) Plan Participation Rates", Journal of Applied Econometrics 11(6), 619-632.

Segerfeldt, F. (2005). "Water for Sale: How Business and The Market Can Resolve the World's Water Crisis", Cato Institute, June 2005.

Shleifer, A. (1998). "State versus Private Ownership", Journal of Economics Perspectives 12(4).

Shi, A. (2000) "How Access to Urban Potable Water and Sewerage Connection Affects Child Mortality" The World Bank, Policy Research Working Paper 2274.

Spiller, P. and W. Savedoff (2000). Spilled Water (eds.), IDB Press. The World Bank. (2002)"Water - the Essence of Life." Development News, May.

Wallsten, S. and Kosec, K. (2005). "Public or Private Drinking Water? The Effects of Ownership and Benchmark Competition on U.S. Water System Regulatory Compliance and Household Water Expenditures" AEI-Brookings Joint Center Working Paper No. 05-05.

Whittington, D., J. Davis, L. Prokopy, K. Komives, R. Thorsten, H. Lukacs, W. Wakeman, and A. Bakalian (2009). "How Well is the Demand-Driven, Community Management Model for Rural Water Supply Systems Doing? Evidence from Bolivia, Peru, and Ghana." Water Policy 11(6).

World Bank (2004). Reforming Infrastructure: privatization, regulation and competition. World Bank and Oxford University Press, Washington, DC.



Figure 1 Map of Uruguay and treated jurisdictions





Figure 3 Average sanitation rate of main cities



Source: ECH and authors' compilations.



Figure 4 Average sanitation rate in treatment and control cities

Source: ECH and authors' compilations.

Figure 5 Sanitation rate in cities with privatized water companies – treatment group



Source: ECH and authors' compilations.

Figure 6 Average count of abnormal levels in treatment and control cities



Source: URSEA and authors' compilations.



Figure 7 Child mortality rate in Uruguay per 1,000 births

Source: Ministry of Public Health and authors' compilations.

Figure 8 Child mortality rate by water related diseases in Uruguay per 1,000 births by treatment and control group



Source: Ministry of Public Health and authors' compilations.

Table 1 Indicators of the public water company: OSE						
•	Units	2002	2003	2004	2005	2006
Service Area						
Total population - water supply	'000 inhabitants	3,158.4	3,178.2	3,100.7	3,245.0	3,253.7
Total population - wastewater	'000 inhabitants	1,791.6	1,805.9	1,541.2	1,919.0	1,935.7
Total number of staff	# FTE	4.816	4.508	4.362	4.174	4.280
Water Service						
Population served - water	'000 inhabitants	3,041.6	3,063.8	2,833.5	2,980.9	3,055.3
Water connections year end	'000 connections	735.1	732.9	737.7	799.9	822.5
Volume of water produced	Million m3/year	282.7	275.1	288.1	309.4	320.0
Total volume of water sold	Million $m3/year$	135.3	131.0	132.3	142.0	147.1
Sewerage Service						
Population served - sewerage	'000 inhabitants	467.8	479.7	531.3	633.8	729.1
Sewerage connections year end	'000 connections	162.2	166.1	169.5	195.6	205.9
Length of sewers	Km	1,683.0	1,759.6	1,872.5	2,410.8	2,494.5
Financial Information						
Total operating revenues	Million USD	137.2	128.3	137.9	176.6	207.8
Total billings to residential customers	Million USD	78.2	72.9	80.8	104.6	123.5
Total billings to industrial customers	Million USD	29.2	26.0	29.0	37.7	45.5
Total water and wastewater operational expenses	Million USD	86.8	79.7	85.3	103.8	121.7
Labor costs	Million USD	46.6	39.5	42.1	47.5	56.1
Total gross fixed assets including work in progress	Million USD	621.8	513.0	542.7	684.5	746.6
Tariff Information						
Fixed charge per month for water and wastewater	USD per month	3.0	3.0	3.3	4.1	4.4
services for residential customers	USD per month					
Connection charges - water	USD	49.1	38.7	41.2	53.2	59.4
Connection charges - sewers	USD	19.7	15.5	16.5	21.3	23.8
Source: OSE						

Relationship be	tween elections & the wa	ter referendum 200	04 (in percentag	(e)		
		National electi	ions & Water	referendum 2004	Council El	ections 2005
Department	City	In favor of	Left Wing	Right Wing	Left Wing	Right Wing
		Nationalization	Main Party	Main Parties	Main Party	Main Parties
Canelones	Canelones	69	54	40	68	26
Canelones	Santa Lucia	75	57	38	65	28
Canelones	Las Piedras	73	58	36	67	27
Canelones	Soca	58	34	61	42	52
Canelones	Pando	68	52	43	57	35
Canelones	Atlantida	65	48	46	58	36
Canelones	Ciudad de la Costa	75	64	29	72	22
Canelones	San José de Carrasco	75	65	28	74	21
Canelones	El Pinar	80	72	21	62	16
Maldonado	Maldonado	54	51	42	48	46
Maldonado	Punta del Este	41	35	56	36	58
Maldonado	San Carlos	56	53	41	55	40
Maldonado	Pan de Azúcar	42	39	55	43	53
Maldonado	Piripolis	52	47	46	47	49
Maldonado	La Barra -Manantiales	55	45	51	41	57
Maldonado	Aiguá	27	22	73	24	72
Rocha	Rocha	74	46	48	50	45
Rocha	La Paloma	76	58	38	63	33
Rocha	Castillos	72	40	53	43	48
Rocha	Lascano	51	34	61	44	51
Rocha	El Chuy	74	55	39	60	34
Lavalleja	Minas	51	36	58	24	72
Lavalleja	Pueblo Solś	36	27	67	16	62
Lavalleja	Pueblo Mariscala	23	18	75	9	86
Lavalleja	José Pedro Varela	47	34	62	22	73
Lavalleja	José Batlle y Ordonez	32	27	70	17	79
Source: Urugua	yan Electoral Court.					

Table 2 Deletionatic Lations of stress 6- the

Table 3 Descriptive statistics for treatment and	d control	groups	in the acces	ss to se	wage ne	twork sa	mple			
			Treated					Control		
Variable	Obs.	Mean	Std. Dev.	Min	Max	Obs.	Mean	Std. Dev.	Min	Max
Network Evacuation Rate	57	0.53	0.12	0.28	0.87	678	0.47	0.20	0.00	0.96
Water Supply Rate	57	0.99	0.01	0.96	1.00	678	0.96	0.06	0.50	1.00
Education (Head of Household)	57	7.10	0.60	6.12	8.31	678	6.95	0.89	3.71	9.89
(log) Household per Capita Income	57	7.78	0.21	7.31	8.09	678	7.55	0.25	6.65	8.25
Unemployment Rate	57	0.13	0.06	0.04	0.25	678	0.11	0.05	0.00	0.32
Poverty Rate	57	0.16	0.09	0.02	0.40	678	0.28	0.13	0.00	0.82
Extreme Poverty Rate	57	0.01	0.01	0.00	0.04	678	0.02	0.03	0.00	0.30
% of Occuppied Households	57	0.16	0.06	0.07	0.31	678	0.14	0.06	0.01	0.33
(log) Accumulated Precipitations	57	6.99	0.19	6.64	7.32	678	7.13	0.27	6.17	7.94
Sources: ECH and the National Meteorc	ology Off	ice.								

Table 3 Descriptive statistics for treatment and control groups in the access to sewage network sample

			Treated					Control		
Variables	Obs.	Mean	Std.	Min	Max	Obs.	Mean	Std.	Min	Max
			Dev.					Dev.		
Fecal coliforms	34	0.00	0.00	0.00	0.00	148	0.41	4.93	0.00	60.00
Pseudomonas aeroginosa	34	0.00	0.00	0.00	0.00	148	0.10	0.30	0.00	1.00
ph	34	7.00	0.42	6.10	8.10	148	7.33	0.48	6.20	8.90
Cloudiness	34	0.97	1.27	0.10	6.06	148	1.50	1.81	0.00	12.60
Count of non-compliance tests	34	0.15	0.36	0.00	1.00	148	0.45	0.79	0.00	4.00
Minimum temperature ($^{\circ}c$)	34	8.76	0.44	7.83	9.30	148	4.80	1.13	2.48	7.38
Average temperature $(^{\circ}c)$	34	16.82	0.16	16.57	17.02	148	17.53	0.97	16.07	20.51
Maximum Temperature (°c)	34	22.18	0.68	20.95	23.05	148	24.06	1.38	21.95	27.35
(log) Accumulated precipitations	34	4.19	0.19	3.82	4.36	148	4.30	0.24	3.50	4.78
Sources: URSEA, ECH and the Nat	ional M	eteorolo	ogy Off	ice.						

Table 4

ce.
Offi
logy
eorol
Mete
nal
atio
le N
d th
I an
ECF
URSEA,
ources:

Table 5 Descriptive statistics by treatment an	d contr	ol grouț	o for th	e child	mortali	ty sam	ple			
			Preated					Control		
Variables	Obs.	Mean	Std.	Min	Max	Obs.	Mean	Std .	Min	Max
			Dev.					Dev.		
Perinatal	35	6.22	8.92	0.00	51.28	140	6.21	6.80	0.00	31.25
Non Perinatal	35	3.43	3.16	0.00	9.55	140	5.59	6.36	0.00	28.99
Total Child Mortality Rate	35	9.65	9.26	0.00	51.28	140	11.80	9.19	0.00	43.69
Child Mortality Rate by Water	35	1.14	1.33	0.00	4.58	140	1.38	2.59	0.00	14.49
Education	35	8.49	0.98	6.99	10.76	140	7.61	1.01	5.70	10.67
Unemployment Rate	35	0.13	0.05	0.05	0.25	140	0.13	0.04	0.05	0.22
(log) Household per Capita Income	35	8.14	0.27	7.72	8.74	140	7.89	0.26	7.13	8.52
(log) Accumulated Precipitations	35	6.98	0.17	6.58	7.22	140	7.05	0.21	6.58	7.29
Minimum Temperature	35	2.23	2.19	-3.10	5.00	140	-0.97	1.77	-4.00	5.00
Sources: Ministry of Public Health,	ECH $\&$	the Na	tional l	Meteor	ology Of	ffice.				

Table 6	;				-	-						
Impact of national	ization c	on access	to sewage	e network	t in Malc	donado ^a		2			ć	
Variables		\mathbf{A}^{p}			ñ			\mathbf{C}^{q}			Ĵ D	
Public Water Provider	$(1) 0.041^{**} (0.016)$	$(2) \\ 0.068^{**} \\ (0.027)$	$(3) \\ 0.041 \\ (0.027)$	$(1) \\ 0.051^{**} \\ (0.022)$	$(2) \\ 0.067^{**} \\ (0.027)$	$(3) \\ 0.039 \\ (0.028)$	$(1) \\ 0.059^{***} \\ (0.012)$	$(2) \\ 0.126^{***} \\ (0.008)$	$(3) \\ 0.099^{***} \\ (0.009)$	$(1) \\ 0.059^{***} \\ (0.014)$	$(2) \\ 0.134^{***} \\ (0.008)$	$(3) 0.114^{***} (0.007)$
Education			0.032^{***} (0.009)			0.030^{***} (0.010)			0.033^{***} (0.009)			0.029^{***} (0.010)
(log) Household per capita income			0.166^{***} (0.028)			0.185^{***} (0.035)			0.156^{***} (0.028)			0.127^{***} (0.040)
(log) Accumulated precipitations			0.051^{***} (0.015)			0.052^{***} (0.019)			0.045^{***} (0.015)			0.041^{**} (0.019)
Time trend by city	No	Y_{es}	${ m Yes}_{ m rec}$	No	Y_{es}	${ m Yes}$	No	Y_{es}	${ m Yes}$	No	Y_{es}	\mathbf{Yes}
Observations Adjusted R2	0.846	635 0.899	0.911	0.851 0.851	580 0.892	0.906 0.906	0.2	702 0.903	0.915	$^{432}_{0.775}$	$^{432}_{0.869}$	$432 \\ 0.881$
Sample Period		1986-2009			1993-2009			1986-2009			1986-2009	
^a Estimates are obtained at city level reported significance at the 5%	cained usin in parent level; ***	ng Papke δ cheses. All [*] stands for	z Wooldric models in significane	lge Fractic clude indi ce at 1% le	onal Logit vidual and evel.	Model. N 1 year effe	larginal eff cts. * stan	ects reporte ds for signi	and robu ficance at	ist standar the 10% le	d errors clı vel; ** staı	istered ids for
^b Treatment Units: Durazno, Florida, Fra Rocha Rosario Río B	Maldonay y Bentos, ranco Sal	do, San Ca Lascano, L lto San Jos	urlos and F ibertad, M té de Mavo	an de Azı elo, Merce San Ram	ícar; Cont ides, Minas ón Santa	rol Units: s, Montevi Lucía, Sar	Artigas, B deo, Paso d andí del Ví	ella Unión, e los Toros, Sarandí Gi	Canelones , Paysandú , Pande	, Carmelo, , Periferia (arembó Tr	Colonia, D Janelones, l andneras, ⁷	olores, Sivera, Freinta
y Tres, Trinidad and ^c Same as in A .	Young.											
^d Only Maldonado ^e Treatment Units: Rivera, Rocha, Salto,	as Treatm Maldonac San José	hent Unit a do; Control de Mayo, 7	nd the sam Units: Art Tacuarembé	e Control tigas, Can 5, Treinta	Units as i elones, Co y Tres and	n A and B lonia, Dura l Trinidad.	azno, Floric	la, Fray Bei	atos, Melo,	Mercedes,	Minas, Pay	sandú,

Table 7	:							2		:		
Impact of nationa.	lization c	n access 1	to sewage	network	in Maldo	nado in	the botto	m 25% of	the inco	me distril	$bution^a$	
Variables	(1)	\mathbf{A}^{b} (2)	(3)	(1)	\mathbf{B}^{c}	(3)	(1)	\mathbf{C}^{d}	(3)	(1)	\mathbf{D}^{ϵ}	(3)
Public water provider	0.076^{***} (0.019)	0.092^{***} (0.026)	0.073^{***} (0.025)	0.080^{***} (0.022)	0.095^{***} (0.027)	0.076^{***} (0.026)	0.109^{***} (0.011)	0.150^{***} (0.008)	0.128^{***} (0.010)	0.118^{***} (0.013)	0.159^{***} (0.010)	0.137^{***} (0.014)
Education			0.017^{**} (0.007)			0.018^{**} (0.009)			0.016^{**} (0.008)			0.012 (0.011)
(log) Household per capita income			0.113^{**} (0.031)			$\begin{array}{c} 0.115^{***} \\ (0.038) \end{array}$			0.090^{***} (0.028)			0.101^{**} (0.033)
(log) Accumulated Precipitations			0.041 (0.034)			0.033 (0.036)			0.028 (0.034)			0.042 (0.040)
Time trend by city	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Observations Adiusted R2	$735 \\ 0.649$	$735 \\ 0.732$	$735 \\ 0.740$	$580 \\ 0.659$	$580 \\ 0.713$	$580 \\ 0.721$	$702 \\ 0.657$	$702 \\ 0.742$	$702 \\ 0.748$	$432 \\ 0.641$	$432 \\ 0.751$	$432 \\ 0.758$
Sample Period		1986-2009			1993-2009			1986-2009			1986-2009	
^a Estimates are ob at city level reported	tained usin in parent	ng Papke $\underline{\delta}$, heses. All	z Wooldrid models inc	ge Fractior clude indivi	al Logit N idual and	Iodel. Ma year effect	urginal effec s. * stands	ts reported s for signifi	l and robu cance at t	st standarc he 10% lev	l errors clu vel; ** staı	ıstered ıds for
significance at the 5% ^b Treatment Units:	6 level; *** Maldonae	[*] stands for do, San Ca	significanc rlos and Pa	e at 1% lev an de Azúc	rel. car; Contrc	ol Units: A	Artigas, Bel	lla Unión, ¹	Canelones,	Carmelo,	Colonia, D	olores,
Durazno, Florida, Fre Rocha, Rosario, Río I	ay Bentos, 3ranco, Sal	Lascano, L lto, San Jos	ibertad, Me é de Mayo,	elo, Merced San Ramó	es, Minas, n, Santa Lı	Montevide ucía, Sarar	eo, Paso de 1dí del Yí, S	los Toros, barandí Gra	Paysandú, inde, Tacué	Periferia C arembó, Tra	anelones, l'anqueras, ^r	livera, Ireinta
y Tres, Trinidad and ^{c} Same as in A .	Young.	tutt of	4+bo com	Control I	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	D Puo V						
OTITA INTERIOR	AS LICAUL	IGTIC OTTIC OT	TIT PITE SETTI		UIII CE STIT	A allu D.						

"Treatment Units: Maldonado; Control Units: Artigas, Canelones, Colonia, Durazno, Florida, Fray Bentos, Melo, Mercedes, Minas, Paysandú, Rivera, Rocha, Salto, San José de Mayo, Tacuarembó, Treinta y Tres and Trinidad.

Table 8												
Impact of nationali	zation o	n access 1	to sewage	network	in Mald	onado in	the top	25% of th	e income	distribut	ion^a	
Variables		\mathbf{A}^{b}			\mathbf{B}^c_{c}			\mathbf{C}^{q}			D^e	
Public water	(1) -0.002	$(2) \\ 0.025$	$(3) \\ 0.001$	$(1) \\ 0.012$	$(2) \\ 0.023$	(3) -0.001	$\substack{(1)\\0.013}$	$(2) 0.081^{***}$	$(3) 0.054^{***}$	$(1) \\ 0.009$	$(2) 0.079^{***}$	(3) 0.064***
provider	(0.028)	(0.034)	(0.036)	(0.033)	(0.033)	(0.036)	(0.015)	(0.011)	(0.010)	(0.012)	(0.007)	(0.006)
Education			0.007 (0.006)			0.004 (0.007)			0.008 (0.006)			0.014^{**} (0.006)
(log) Household per capita income			0.214^{***} (0.037)			0.232^{***} (0.042)			0.220^{***} (0.037)			0.151^{***} (0.046)
(log) Accumulated precipitations			0.052^{**} (0.024)			0.055^{*} (0.031)			0.052^{**} (0.024)			0.043^{***} (0.009)
Time trend by city	No	$\mathbf{Y}_{\mathbf{es}}$	Yes	No	Yes	Yes	N_{O}	\mathbf{Yes}	Yes	No	Yes	Yes
Observations	735	735	735	580	580	580	702	702	702	432	432	432
Adjusted R2	0.873	0.901	0.912	0.871	0.894	0.907	0.876	0.905	0.917	0.719	0.794	0.816
Sample Period		1986-200	6		1993-2009	6		1986-2009			1986-2009	
^a Estimates are obt	ained usin	ig Papke &	Wooldrid	ge Fractio	nal Logit	Model. M	arginal eff	ects reporte	ed and robu	ist standa	rd errors ch	istered
at city level reported significance at the 5%	ın parenu level; ***	stands for	models inc significanc	e at 1% le	rauar and vel.	year enec	us. stan	Ids IOF SIGN	incance at 1	tue 10%0 16	evel; ** stal	Ids Ior
^b Treatment Units:	Maldonac	lo, San Ca	rlos and Pa	an de Azú	car; Conti	rol Units:	Artigas, B	sella Unión,	Canelones	, Carmelo,	Colonia, D	olores,
Durazno, Florida, Fray Rocha, Rosario, Río Bi	· Bentos, J :anco, Sali	Lascano, L to, San Jos	ibertad, Me é de Mayo,	elo, Mercec San Ramć	tes, Minas in, Santa I	i, Montevic Jucía, Sara	ieo, Paso c mdí del Yí,	le los Toros , Sarandí Gi	, Paysandú, rande, Tacu	Periferia arembó, T	Canelones, l ranqueras, ^r	Aivera, Ireinta
y Ires, Irinidad and Y ^c Same as in A .	oung.											
^d Only Maldonado a	s Treatme	ent Unit ar	id the same	e Control ¹	Units as ir	n A and B.						
^e Treatment Units: Rivera, Rocha, Salto, 9	Maldonad San José c	o; Control le Mayo, T	Units: Art acuarembó	igas, Cane , Treinta y	lones, Col ^r Tres and	onia, Dura Trinidad.	zno, Floric	la, Fray Be	ntos, Melo,	Mercedes,	Minas, Pay	sandú,

Variables	Pooled	Pooled	Within	Within
	FGLS	OLS	Estimator	Estimator
				D-K
Public Water Provider	0.067^{***}	0.104^{***}	0.097^{***}	0.115^{***}
	(0.019)	(0.032)	(0.033)	(0.030)
Education	0.024***	0.021***	0.019^{**}	0.031^{**}
	(0.003)	(0.008)	(0.008)	(0.011)
(log) Household per Capita Income	0.154^{***}	0.156^{***}	0.161^{***}	0.133^{***}
	(0.013)	(0.035)	(0.035)	(0.038)
(log) Accumulated Precipitations	0.012*	0.029	0.023	0.043^{*}
	(0.007)	(0.019)	(0.016)	(0.022)
Observations	432	432	414	432
AB(1)	0.366	0.270	0.397	_
Sample Period	0.000	198	6-2009	

Impact of nationalization on access to sewage network controlling for AR(1) disturbances^{*a*}

Table 9

^aAll columns report estimates using panel data estimators controlling for AR(1) disturbances. Marginal effects reported and robust standard errors clustered at city level reported in parentheses. All models include individual and year effects, as well as time trends by city. Treatment Units: Maldonado; Control Units: Artigas, Canelones, Colonia, Durazno, Florida, Fray Bentos, Melo, Mercedes, Minas, Paysandú, Rivera, Rocha, Salto, San José de Mayo, Tacuarembó, Treinta y Tres and Trinidad. * stands for significance at the 10% level; ** stands for significance at the 5% level; *** stands for significance at 1% level.

	Poi	sson	Negative	Binomial
Variables	(1)	(2)	(1)	(2)
Public water provider	-0.650*	-0.693*	-0.650*	-0.694*
	(0.358)	(0.367)	(0.355)	(0.384)
Precipitations		0.381		0.381
		(0.381)		(0.386)
Minimum temperature		-0.098		-0.098
		(0.121)		(0.122)
Average temperature		0.001		0.001
		(0.152)		(0.152)
Individual Effects	Yes	Yes	Yes	Yes
Year Effects	Yes	Yes	Yes	Yes
Observations	182	182	182	182
Log Likelihood	-101.316	-100.954	-101.316	-100.954
Sample		$2004 \cdot$	- 2009	

Table 10 Impact of nationalization on water quality^a (abnormal levels of microbiological elements)

^{*a*}All columns report estimates using a zero inflated model. Marginal effects reported and robust standard errors clustered at city level reported in parentheses. All models include individual and year effects. Vuong tests indicated that the zero inflated models are appropriate. * stands for significance at the 10% level; ** stands for significance at the 5% level; *** stands for significance at 1% level.

Impact of nation	lization (on child	mortalit	y by wate	r-related	$diseases^a$						
				All					Mal	donado		
Variables	Poiss	uc	Negative	Binomial	Papke & V	Wooldridge	Pois	son	Negative	Binomial	Papke & W	'ooldridge
Public water	(1) -0.632	(2)	(1)	(2) -0.031	(1) -0 001**	(2) -0.000	(1)-0.636	(2) 0.061	(1)	(2)	(1) - 0.001**	(2) -0.000
provider	(0.403)	(0.581)	(0.403)	(0.581)	(0.000)	(0.001)	(0.440)	(0.680)	(0.440)	(0.680)	(0.00)	(0.001)
Education		-0.158		-0.158		0.000		0.008		0.008		0.000
	-	(0.431)		(0.431)		(0.001)		(0.477)		(0.477)		(0.001)
Unemployment	1	11.649^{*}		11.649^{*}		0.014		11.371		11.371		0.015
rate	-	(6.972)		(6.972)		(0.00)		(7.161)		(7.161)		(0.009)
(log) Household		0.677		0.677		0.001		0.323		0.323		0.001
per capita income	-	(1.343)		(1.343)		(0.002)		(1.507)		(1.507)		(0.002)
(log) Accumulated		-1.915		-1.916		-0.001		-1.934		-1.934		-0.001
precipitations	-	(2.349)		(2.349)		(0.004)		(2.379)		(2.379)		(0.004)
Minimum		0.047		0.047		-0.000		0.020		0.020		-0.000
temperature	-	(0.247)		(0.247)		(0.000)		(0.248)		(0.248)		(0.000)
Observations	133	133	133	133	175	175	126	126	126	126	168	168
Log Likelihood	-96.314 -	-93.978	-96.314	-93.978	-1.445	-1.438	-88.902	-86.856	-88.902	-86.856	-1.395	-1.388
^a Marginal effects and negative binomi	reported. al estimate:	All model s report co	ls include ; onventiona	year effects I standard e	$\&$ judicial \vdots	section fixed renthesis, wh	effects. Th iile the Pap	ne data are ke & Wool	e for the pe dridge esti	eriod 2003-2 mates repor	2009. The Point robust sta	oisson ndard
errors clustered at t. is appropriate in thi 1% level.	ne city leve s sample.	el in paren * stands f	tthesis. 'L'h for signific.	ie Vuong te. ance at the	st suggest t 10% level;	hat the stan ** stands fo	dard poiss or significa:	on model, nce at the	and not th 5% level;	e zero-inflat *** stands f	ted poisson for significa	model ace at

Table 11 Immost of notionalization of